Massachusetts Institute of Technology Handout 12 6.042J/18.062J: Mathematics for Computer Science October 3, 1996 Professors Nancy Lynch and Tom Leighton

Quiz 1 Information

The quiz will be HARD!

When: Wednesday, October 16, 7:00 to 9:00 PM Students with conflicts only: 9:00 to 11:00 PM

(NOTE: Students who take the quiz 7:00 to 9:00 PM, MUST stay until 9:00 PM; students who take it 9:00 to 11:00 PM, MUST enter the exam room by 9:00 PM.)

Where: 54-100

Crib Sheet: The quiz is not open-book, but you are allowed to bring one 2-sided sheet to the exam. You may have whatever information you like on it but it must be handwritten and must not be xeroxed.

Syllabus: Everything done until and including the recitation on October 11.

Make-up Exam: If you have a conflict come see us NOW.

Special office hours: There will be special office hours during quiz-week as follows: Mojdeh : T uesday, October15, 9 � 11 Alex : T uesday, October15, 4 � 6 Dimitri : T uesday, October15, 5 � 7 Gunnar : W ednesday, October16, 2 � 4 Y ong Y eow : W ednesday, October16, 4 � 6